

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Giovannetti, Giorgia; Sanfilippo, Marco; Vivoli, Arianna

Working Paper Trade liberalization, employment, and gender in Ethiopia

WIDER Working Paper, No. 2021/59

Provided in Cooperation with: United Nations University (UNU), World Institute for Development Economics Research (WIDER)

Suggested Citation: Giovannetti, Giorgia; Sanfilippo, Marco; Vivoli, Arianna (2021) : Trade liberalization, employment, and gender in Ethiopia, WIDER Working Paper, No. 2021/59, ISBN 978-92-9256-997-6, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki, https://doi.org/10.35188/UNU-WIDER/2021/997-6

This Version is available at: https://hdl.handle.net/10419/243385

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WIDER Working Paper 2021/59

Trade liberalization, employment, and gender in Ethiopia

Giorgia Giovannetti,^{1,2} Marco Sanfilippo,^{3,4} and Arianna Vivoli^{1,*}

March 2021

United Nations University World Institute for Development Economics Research

wider.unu.edu

Abstract: This paper analyses the impact of trade liberalization on local labour markets in Ethiopia, with a focus on the gender dimension of employment. By exploiting rich micro-level data on Ethiopian workers, we evaluate the effect of the Ethiopian trade reforms on the changes and composition of employment, adopting as unit of analysis Ethiopian districts. We find that districts more exposed to trade liberalization experienced reductions in their employment levels, especially in female employment. We also show that reductions in (agricultural) input tariffs triggers a process of sectoral reallocation from agriculture to services and that this process is particularly pronounced for women. This in turns contributes to increase sectoral segregation.

Key words: tariffs, employment, structural transformation, gender, Ethiopia

JEL classification: C23, F14, F16, J16

Acknowledgements: Access to the individual survey data used in this paper has been granted under a confidential agreement with the Central Statistics Agency of Ethiopia. The authors thank Fitsum Mulugeta and the Ethiopian Economic Association for facilitating access to the data.

This study has been prepared within the UNU-WIDER project Women's work - routes to economic and social empowerment.

Copyright © UNU-WIDER 2021

UNU-WIDER employs a fair use policy for reasonable reproduction of UNU-WIDER copyrighted content—such as the reproduction of a table or a figure, and/or text not exceeding 400 words—with due acknowledgement of the original source, without requiring explicit permission from the copyright holder.

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9256-997-6

https://doi.org/10.35188/UNU-WIDER/2021/997-6

Typescript prepared by Ayesha Chari.

United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland, Sweden, and the United Kingdom as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

¹ University of Florence, Firenze, Italy; ² European University Institute, Firenze, Italy; ³ University of Turin, Torino, Italy; ⁴ University of Antwerp, Antwerpen, Belgium; * corresponding author: Arianna Vivoli, arianna.vivoli@unitn.it

1 Introduction

In the 1980s, several developing countries started to implement trade reforms to stimulate their economies and possibly improve their living standards. Ethiopia was no exception to this general trend. Following the Structural Adjustment Programme from the World Bank and the International Monetary Fund (Easterly 2003), from 1992–93 onwards Ethiopia implemented trade reforms in order to reduce barriers to trade and to open up to foreign competition as a strategy of long-term growth (Jones et al. 2011). However, evidence on the effects of trade liberalization in Ethiopia is scant and mostly focused on (manufacturing) firms (Bigsten et al. 2016; Fiorini et al. 2019).

In this paper, we use micro-level data to investigate the effects of trade liberalization on local labour markets in Ethiopia. More specifically, we exploit variation in output and input tariffs across local labour markets over time to check whether and how increased international competition affects employment. In addition, with Ethiopia being largely characterized by high levels of agricultural employment, we ask specifically whether trade liberalization has affected the process of structural transformation in the country.

Since tariffs vary across goods, and households have different sources of income and spending habits, trade policies are likely to have a heterogeneous impact on workers in different sectors, with different skills, and across women and men. Throughout the paper we adopt a gender perspective. Whereas tariff reductions took place predominantly in female-dominated sectors, in the period covered by our study, the gender employment gap dropped from 20 percentage points in 1994 to 12 percentage points in 2013. The structural transformation of the Ethiopian economy from the agricultural sector to the services sector seems to concern especially female workers, but with only low-skilled services growing following a reduction in input tariffs. Understanding whether trade liberalization has fostered gender-specific changes in the labour market is an important issue in view of the large gender gaps that remain in the Ethiopian labour market (Essers et al. 2021; WB and WTO 2020).

Theoretically, changes in tariffs result in two simultaneous effects: on the one hand, trade liberalization increases competitive pressures in the output market; on the other hand, it also increases the competitiveness of firms by reducing their input costs.

According to the literature, a reduction in output tariffs due to trade liberalization may stimulate labour demand through two opposite channels (Winters and Di Ubaldo 2020).¹ It can generate disruptive effects on employment levels since domestic markets may be penetrated by cheaper imported products. The negative price shock can, in turn, hamper domestic production, thereby inducing firms to fire workers if their production drops or if they exit the market [this was found, for instance, by Revenga (1997) for Mexico and by Dix-Carneiro and Kovak (2017) for Brazil]. On the other hand, the pro-competitive effect of a drop in output tariffs may induce firms to invest to remain competitive; if these investments bring factor substitution towards labour, international trade can also increase labour demand (e.g., for Mauritius: Milner and Wright 1998).

For input tariffs, theory predicts different effects. Lower input tariffs are supposed to increase firms' productivity through several channels, including via a learning effect, through the acquisition

¹ For an extensive review on the subject, see Goldberg and Pavcnik (2007b), Hoekman and Winters (2005), and Cirera et al. (2014).

of foreign technology embodied in imported inputs (in some cases, leading to the so-called skillbiased technological change), through an increase in the quality, and the variety of inputs. The effect of lower input tariffs in low-income countries has been empirically investigated, among others, by Amiti and Konings (2007) and Goldberg et al. (2010) for Indonesia and India, respectively. Both studies find that cheaper inputs (because of lower input tariffs) increase firms' productivity. Goldberg et al. (2010) also find an increase in new products, output, and research and development for Indian firms. This increase in productivity might expand production and, therefore, increase employment. At the same time, if productivity growth exceeds output growth (for instance, because of the lack of skilled workers or infrastructure and because in some sectors technological inputs free up workforce that is not absorbed by other sectors), employment might also decrease (Meschi and Vivarelli 2009).

Trade liberalization may have relevant heterogenous consequences across genders; women tend to hold a disproportionate share of low-skilled jobs and often are engaged in less productive activities, especially in agriculture (WB and WTO 2020). Hence, trade policy, which is in principle gender neutral, can affect women more. The literature has proposed several channels through which trade liberalization may affect female labour force participation. The first is by reducing discrimination. As initially proposed by Becker (1957), only employers with market power can engage in discriminatory practices, whereas employers operating in a perfectly competitive market cannot, because discrimination is costly. Becker's theory, confirmed by many empirical tests (e.g., for industries in the United States: Black and Brainerd 2004; for Colombia: Ederington et al. 2009), implies that women should benefit more from an increase in competition (i.e. from a lowering in output tariffs), thereby lowering the gender employment gap. In a similar vein, Schultz (2007) highlights that trade liberalization triggers an increase of human capital and gender equality. Another channel through which output tariffs might affect women has to do with social norms. Indeed, if cultural norms, preferences or even stigmas play a role in the definition of which sectors are more 'appropriate' for men and women, when foreign competition induces structural transformation, there might be gender-specific shifts in the reallocation across sectors, changing the sectoral segregation of the economy (Goldin 1995; Gaddis and Pieters 2017; Wang et al. 2020). Further, given that technologies can affect the composition of employment, a shift towards more capital or technological-intensive activities can favour employment of women because women are less likely to be involved in physically demanding activities than men. If women's labour productivity is more closely tied to the capital intensity of production, the adoption of more sophisticated technologies in response to input or output tariff reductions can favour gender employment and their wages [as shown for instance by Juhn et al. (2014) for Mexico]. Finally, provided that a certain level of sectoral segregation of male and female workers is common, the outcomes of trade liberalization in terms of structural transformation will not be neutral to existing structures (Pieters 2018). For instance, in Indonesia input tariff reductions in the 1990s led to higher employment rates, which happened to be concentrated among women, since femaleintensive industries relied to a greater extent on inputs for which import tariffs declined most (Kis-Katos et al. 2018).

Ultimately, the impact of trade policies, as well as their consequences on gender and structural transformation, is an empirical question, possibly different in different countries, and between different groups of workers and sectors. We address this question using micro-level data for Ethiopia over a 20-year period. In our study, we combine the 1994 population census with the three waves of the Ethiopian National Labour Force (NLF) Survey, covering the years 1999, 2005, and 2013 (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014). Individual-level information is re-aggregated in a panel dataset of 388 districts (*woreda*, the third administrative division of Ethiopia). Our identification strategy links changes in tariffs with changes in the

composition of the local workforce on the basis of a district's exposure to trade (as in Autor et al. 2016; Kovak 2013; Topalova 2007).

Our results show that districts more exposed to trade liberalization experienced a reduction in employment—a result that is robust to several specifications—and that female workers were more affected than male workers. Moreover, we find evidence of a contribution of trade liberalization (especially in the form of lower input tariffs) to the structural transformation of the Ethiopian economy from agriculture to services, even though the shift to services cannot make up for the employment losses in the agricultural sector; this transition seems to be mostly driven by a higher mobility of the female component of the workforce. In this context, if stringent social norms or preferences are in place, liberalization and sectoral reallocation can have important implications for women, increasing sectoral segregation. We test this channel and find that input tariffs increase sectoral segregation. Finally, we extend our analysis to other dimensions related to employment, by looking at the effect of trade liberalization on unemployment, education, and movements within the services sector; we find that reductions in input tariffs decrease women's educational attainment (but not for men) and push them to move into low value-added industries, such as community services and private household activities.

Our work fits into a large body of literature that has investigated the effects of trade liberalization on employment and gender. Although initially this literature has mostly focused on the firm level (e.g., Edwards 2004) or on the worker level (e.g., for Mexico: Feliciano 2001; for Colombia: Attanasio et al. 2004), more recent work has looked into labour markets with a regional perspective. To some extent, our findings are consistent with the results from various other contexts in which trade liberalization has been found to reduce employment (e.g., for India: Topalova 2007; for Brazil: Kovak 2013; Dix-Carneiro and Kovak 2017; for the United States: Autor et al. 2016; for South Africa: Lepelle and Edwards 2020).

Our findings on the gender-specific effects of trade liberalization contributes to the existing evidence, which so far does not provide univocal results. Wang et al. (2020) and Kyander (2020) find positive effects of import competition on female work for China and India, respectively, especially in the manufacturing sector and for women with limited or no schooling. Trade liberalization (proxied by a decrease in output tariffs) in Mexico² increases formalization in the manufacturing sector between genders, but men benefit the most, while in the services sector, women experience an increase in informality (Ben Yahmed and Bombarda 2020). More in line with (some of) our results, female workers in Indonesia have responded more strongly to the positive labour demand shock of input tariff liberalization than male workers (Kis-Katos et al. 2018). Similarly, in Brazil lower output tariffs reduce the gender employment gap and increase male unemployment, also shifting general employment towards trade and services (Gaddis and Pieters 2017). A similar shift in our analysis is due to input rather than output tariffs.

Our results are also consistent with international trade literature showing that openness to trade can shape the process of structural transformation (Fajgelbaum and Redding 2014). In this respect, Erten et al. (2019) suggest the existence of a causal relationship between trade openness and structural transformation away from agriculture across Chinese counties following WTO accession. For South Africa, Lepelle and Edwards (2020) find that tariff liberalization is associated

 $^{^{2}}$ A recent report (WB and WTO 2020) points out that the impact of trade policy on the probability of working formally for women (in Mexico after the North American Free Trade Agreement) are not uniform (positive for large firms and negative for small firms), but also that the African Continental Free Trade Area is likely to boost female employment, particularly for skilled women, with the gender or skill wage gap worsening in only a few countries.

with a decline in the manufacturing-services ratio, even though their result is driven by a decline in the manufacturing sector rather than by a rise in services, especially in more exposed regions. Among other things, they also find that the decline in manufacturing employment was more pronounced for Black women, reflecting their relevant presence in those sectors more exposed to tariff cuts (e.g., clothing and textiles) and that also this group was unable to relocate into the services sector.

The remainder of the paper is structured as follows. Section 2 provides a quick overview on the trade liberalization process of Ethiopia over the period 1993–2013. Section 3 describes the data and methodology. Section 4 contains our main results. The underlying mechanism and extensions are analysed in Section 5 and Section 6 concludes.

2 Ethiopian trade liberalization

By the beginning of the 1990s, like many other developing countries, Ethiopia abandoned the import-substituting policy that the country had pursued both in a private-led way (from the early 1950s to 1974, during the Imperial regime) and in a public-led way (from 1974 to 1991, under the Dergue regime) (Gebreeyesus 2016). In 1991 the new government, led by the Ethiopian People's Revolutionary Democratic Front, decided to turn to a more market-oriented economic policy and to implement various reforms under the Structural Adjustment Programme sponsored by the World Bank and the International Monetary Fund. As far as trade policy was concerned, the main objectives were the reduction of both tariff and non-tariff barriers and the devaluation of the *birr*, the national currency. The Ethiopian government implemented six waves of tariff reforms between 1993 and 2003; before 1993 the tariff rates ranged from 0 to 240 per cent, in 1995 the range was reduced to 0–80 per cent, and in 2002 to 0–35 per cent; meanwhile, the number of tariff bands fell from 23 to 6, including the 0 tariff band (Bigsten et al. 2016).

In this paper, we use tariffs data from the World Integrated Trade Solution (WITS) database (see World Bank 2021). More precisely, we extract the Ethiopian import tariff rates³ at the three-digit level (following the ISIC Rev. 3 classification; see United Nations 1990) for each of the years in our dataset and use them to construct our trade exposure measures for both input and output tariffs. Following Topalova (2007, 2010), Ethiopian industry-specific import tariffs are weighted using the district-level composition of employment at the baseline year.⁴

For output tariffs, we calculate the number of employed workers in each sector s at the baseline $(Empl_{s,k,1994})$ by district k and divide these figures for the total employment by district $(Empl_{k,1994})$.⁵ We then merge these weights with time-varying sector-specific tariff data

³ We use the trade-weighted average of the World Trade Organization's 'Most Favoured Nation' *ad valorem* tariff rate at the industry level.

⁴ To be more in line with Topalova (2007), we should have used pre-liberalization information on employment in districts; that is, in the Ethiopian case, information before 1993. Unfortunately, our most remote information comes from the 1994 census, which is 1 year after the starting of the liberalization period. We acknowledge this limit of our data.

⁵ Given that tariff data from the WITS database are expressed using the ISIC Rev. 3 at the three-digit level, we harmonized the sectoral information from the 1994 census and the 1999, 2005, and 2013 NLF to this international classification.

(*Tarif f*_{*s*,*t*}). Lastly, we sum by district. In this way, we end up having a district-level trade exposure measure, as shown in the formula below:

$$Output Tariff_{k,t} = \sum_{s=1}^{S} \frac{Empl_{s,k,1994}}{Empl_{k,1994}} \times Tariff_{s,t}$$

where t=1994, 1999, 2005, and 2013. This measure indicates how much a district has been exposed to output tariff reduction, given the district's employment composition across sectors in the initial year. For the way it is constructed, a reduction in output tariffs should capture the effect of a reduction of prices in the output market and, therefore, an increase in foreign competition. As far as the non-tradeable sector⁶ is concerned, we follow Autor et al. (2016), Kis-Katos and Sparrow (2011), and Topalova (2007) and we treat it as an additional sector, setting tariffs to zero, thereby including also its workforce in the weighting of tariffs.⁷

To account for input tariffs, we create an analogous district-level trade exposure measure:

Input
$$Tariff_{k,t} = \sum_{s=1}^{S} \left(\frac{Empl_{s,k,1994}}{Empl_{k,1994}} \times \sum_{i=1}^{I} \left(\frac{M_{i,s,2005}}{M_{s,2005}} \times Tariff_{s,t} \right) \right),$$

which is equivalent to the output tariff exposure measure, with the only difference that here tariffs are first weighted with the input shares $M_{i,s}$ that each sector *s* acquires from each sector *I*, and then the product is weighted for the district's employment share. We extract information on input shares from the 2005–06 Ethiopia Input Output (I/O) Table and Regionalized Social Accounting Matrix (SAM), constructed by the Ethiopian Development Research Institute and the International Food Policy Research Institute (IFPRI) (see Tebekew et al. 2009).⁸ To construct the input tariff measure we proceed in the following way: as the Ethiopian SAM table we used is only commodity-by-activity (with 58 commodities and 47 activities⁹), and only the activity account is classified according to the ISIC Rev. 3.1 classification, we first had to harmonize the commodity account according to the ISIC classification and assign each commodity to an activity category; once done, it is possible to assign the tariff level of the same ISIC class to each category (coded according to the ISIC classification). Next, we calculated the industry-specific input tariffs for each year of our panel by multiplying each I/O coefficient (from the 'intermediate input demand' section of the I/O table) for its tariff and, finally, we weighted them for the district industryspecific employment weights, as for the output tariffs.

⁶ We classify as non-tradeable sectors all those industries for which the WITS database does not provide information on tariffs. This includes all industries included from division E to division Q in the ISIC Rev. 3 classification.

 $^{^{7}}$ We also perform the baseline analysis for output tariffs excluding the non-tradeable sector from the weights. Estimates are still significant and point to the same direction, but figures get smaller. Conversely, for the input tariff analysis we follow Kis-Katos and Sparrow (2015) and Kis-Katos et al. (2018) and we include both tradeables and non-tradaebles in the output sectors (*s*), but only tradaebles in the input suppling sectors (*i*).

⁸ We acknowledge that the optimal solution would have been to use data from a pre-sample input–output (I/O) table; unfortunately, the IFPRI I/O table is the first available constructed with real Ethiopian data. Moreover, we are quite confident that I/O coefficients remain quite stable over time.

⁹ Usually the number of activities in a social accounting matrix (SAM) tends to be smaller, but the IFPRI SAM has a relatively high level of disaggregation for production activities and commodities in the Ethiopian economy, especially for what concerns the agricultural sub-sectors.

Figure 1 shows changes of average (input and output) tariff rates in Ethiopia over the period covered in this study.¹⁰ Figure 2 shows how output and input tariff exposure measure varies substantially across districts over the period considered.



Figure 1: Trends in tariffs

Source: authors' elaboration on WITS data (World Bank 2021).



Figure 2: Changes in output and input tariffs (%) by Ethiopian districts

Source: authors' elaboration on the census, NLF, and WITS data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014; World Bank 2021).

Figure 3 reports the percentage change in tariff reduction for the main economic sectors, defined according to the two-digit code of the ISIC Rev. 3 classification. It shows that the larger reductions took place in those sectors that absorb most of the Ethiopian workforce, namely agriculture (-75

¹⁰ Unfortunately, WITS reports tariffs for Ethiopia only since 1995 and not for all years (data on tariffs are missing for 1996–2000, 2003–05, 2007, and 2013). Therefore, we use the 1995 tariff rate as a proxy for the 1994 tariff rate and we linearly interpolate tariff data where missing.

per cent), food production (-68 per cent), and wood (-72 per cent), whereas the only sector more protected was the basic metal sector (+74 per cent).





Source: authors' elaboration on WITS data (World Bank 2021).

3 Data and empirical analysis

3.1 Worker-level data

We harmonize and merge the 1994 population census with three existing waves of the Ethiopian NLF Survey, for 1999, 2005, and 2013. The 1994 census provides information on the sociodemographic characteristics of the respondents. It provides information also on their employment status, including the sectoral distribution of workers.¹¹ The NLF, administered by the Central Statistics Agency (CSA),¹² provides information about labour under-utilization, unemployment, and the working conditions of persons in employment. We restrict the sample to the working age population (i.e. those aged 15–64 years¹³) and harmonize the sectoral affiliation of workers using the ISIC Rev. 3 classification.

¹¹ Information on the employment status of individuals, and their sectoral distribution, is not included in the most recent census (2007).

¹² The NLF surveys are nationally representative. They cover all urban and rural areas of the country, except the nonsedentary areas in the Somali region. The sampling frame to select enumerator areas is provided by the population census (the 1994 census for the 1999 and 2005 NLF and the 2007 census for the 2013 NLF). All the relevant information on the sampling procedures, coverage, and full descriptive statistics are available from the survey reports published by the Ethiopian Central Statistical Agency (see CSA 2014).

¹³ We also check the robustness of our results by running the baseline regressions with the whole sample (i.e. including workers not only between 15 and 64 years but of all ages) and the results remain stable.

After harmonizing the 1994 census with the three waves of NLF, we collapse the individual-level dataset to the district level (*woreda*) using sampling weights. The result is a panel of Ethiopian districts over 4 years. For consistency, we decided to confine the analysis only to *woreda* whose name appear at least twice in the four waves of the panel. This results in an unbalanced panel including 388 districts representing the 88.71 per cent of estimated population and 89.31 per cent of the total workforce.

Table 1 provides an overview on the socio-demographic characteristics of our working sample. Figure 4 provides change in the distribution of male and female workers across sectors between 1994 and 2013. Employment levels are generally high while the employment gender gap has decreased from almost 20 percentage points in 1994 to 12 percentage points in 2013.¹⁴ Education steadily increases but the educational attainments remain quite low, especially for women (on average, in 2013 women have 1.98 years of education whereas men have 3.31 years).

	1994	1999	2005	2013
Male (%)	49.13	47.40	47.18	48.86
Age (years)	31.38	31.83	31.67	31.75
Urban (%)	9.59	12.86	13.75	5.52
Employed (%)	78.50	64.06	84.07	82.25
Unemployed (%)	1.12	1.54	1.34	1.09
Years of school	0.95	1.21	1.70	2.62
Illiterate (%)	78.19	69.73	63.49	49.40
Agriculture (%)	88.89	72.96	76.31	74.40
Manufacturing (%)	1.78	6.19	7.36	6.25
Services (%)	6.68	17.96	14.89	18.49

Table 1: Socio-demographic characteristics of the working sample

Source: authors' elaboration on the census and NLF data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014).

Needless to say, the predominant source of employment is agriculture, especially among men. While manufacturing was not yet able to take off in the Ethiopian economy,¹⁵ structural transformation is going towards the services sector, which in 2013 employed almost 21 per cent of the total workforce. In proportion, female employment reduced more than that of men in the agricultural sector and increased more in the services sector.

¹⁴ The employment rate of men (women) is calculated by dividing the number of men (women) aged 15–64 years in employment by the total male (female) population of the same age group.

¹⁵ The weaknesses of the manufacturing sector in Ethiopia has been widely recognized by many studies (Diao et al. 2021). To give an idea, Sutton and Kellow (2010), who conducted an in-depth analysis of the Ethiopian secondary sector, documented that 78 per cent of exports come from a few labour-intensive primary sector industries: coffee, oilseeds, khat (chat), while the remaining 22 per cent come from only four industries: cut flowers, leather, meat and meat products, and clothing and textiles.



Figure 4: Change in distribution of workers across sectors in 1994 and 2013

Source: authors' elaboration on the census and NLF data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014).

In addition to the two aforementioned gender-related stylized facts, that is, a decreasing employment gender gap and a (though not yet well pronounced) shift from the male-dominated agricultural sector to the female-dominated services sector, there is a third fact that motivates our focus on gender. Figure 5 plots changes in tariffs against the initial share of female employment in each industry. Sectors that in 1994 were more female-intensive witnessed larger tariff cuts, especially output tariff cuts, suggesting that we might find differential impacts of tariffs reduction across gender.



Figure 5: Tariff declines (1994-2013) and initial female employment shares (1994)

Source: authors' elaboration on the census, NLF, and WITS data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014; World Bank 2021).

3.2 Empirical strategy

We aim at estimating the impact of trade liberalization on employment levels by adopting a local labour market perspective and by considering the Ethiopian trade liberalization as a sort of natural experiment. This is possible considering the relatively short-time frame (1993–2003) in which Ethiopian tariffs were reduced and, as we will argue later, the likely exogenous nature of this policy. The identification strategy follows existing work on the effect of trade liberalization on local labour markets (Autor et al. 2016; Kovak 2013). By exploiting the cross-sectional variation in the employment structure of districts at the onset of the trade liberalization programme, we aim at comparing the heterogenous response in terms of district employment levels and other socio-economic outcomes between districts more exposed to tariff reduction compared with those more protected.

For the causal identification to be valid, the treatment (i.e., the exposure to trade liberalization) should be exogenous to the outcomes analysed. There is a widespread concern in the trade literature regarding the endogeneity of tariffs; indeed, it is reasonable to question whether labour market conditions, such as a fall in the employment rate for instance, might influence political decisions about raising tariffs (or taking other protectionist measures) in some sectors, thereby introducing a bias in our estimates. The issue has been widely explored in both the theoretical and empirical literature (Grossman and Helpman 1994) and scholars have recognized the potential interference of industrial lobbies and interest groups in the delineation of trade policies.

We acknowledge that, for their intrinsic characteristics, trade policies will always show some degree of endogeneity. However, this is not necessarily a concern in our study. Indeed, Jones et al. (2011) analyse tariff reforms of four countries in Eastern Africa, including Ethiopia, between the early 1990s and early 2000s; they conclude that there was limited evidence of political distortions in the tariff reduction process in these countries, mainly because the technocratic reforms were implemented according to World Bank guidelines, leaving little room for lobbyist pressures.¹⁶ As in Jones et al. (2011), in Appendix Table A1 we report some descriptive statistics for output and input tariffs throughout the period under study, to better understand the process behind the Ethiopian tariffs reform. As expected, tariffs decreased on average by almost 10 percentage points, but also their dispersion decreased, as signalled by a decreasing standard deviation. Moreover, we can also see how both the kurtosis and the skewness approach zero over time, implying respectively that the distribution gained more symmetry around the mean and lost weight in its tails; that is, there were fewer extreme cases. Consistent with this trend, in Figure 6 we can see how in Ethiopia the higher the tariffs around the baseline, the higher the reductions in targeted sectors. Next, as in Topalova (2007), we regress tariffs against industrial characteristics to test whether trade policy has been used to protect particular sectors. We did this by aggregating our data at the three-digit sectoral level; results are reported in Table 2. Specifically, we regress the change in tariffs (output tariffs Columns 1-3, input tariffs Columns 4-6) on sectoral employment at the baseline (a large sectoral workforce might translate into a large electoral basis if protected) and on the share of illiterate workers (as a proxy for unskilled workers, likely to be the most vulnerable to trade liberalization). The only significant relationship appears to be the one with initial levels of tariffs, suggesting that tariff reductions were faster for those sectors that were more protected at the baseline. This should imply, cautiously, that tariff rate reductions are part of a generalized trend towards liberalization.

¹⁶ This was also the case for many other developing countries in the 1990s as well, such as Colombia's tariff reduction for WTO accession (Goldberg and Pavenik 2007a, b) or Brazil's liberalization for the Mercosur (Dix-Carneiro and Kovak 2017).



Figure 6: Changes in output and input tariffs (%) compared with the baseline

Source: authors' elaboration on WITS data and IFPRI SAM (World Bank 2021; Tebekew et al. 2009).

	(1)	(2)	(3)	(4)	(5)	(6)
	∆ Output tariffs	∆ Output tariffs	∆ Output tariffs	∆ Input tariffs	∆ Input tariffs	∆ Input tariffs
Employment	-7.970			-5.889		
	(6.538)			(4.245)		
Share of illiterate workers		-0.151			0.198	
		(0.872)			(0.174)	
Baseline tariffs			-0.192***			-0.144***
			(0.018)			(0.017)
Observations	133	124	195	336	309	330
R-squared	0.011	0.000	0.366	0.006	0.004	0.186

Table 2: Tariff declines and baseline industrial characteristics

Note: *p<0.1, **p<0.05, ***p<0.01. Standard errors are in parentheses. Employment rate is calculated by dividing the number of people aged 15–64 years in employment by the total population of the same age group in the same sector. The sectoral share of illiterate workers is calculated over the number of employed people aged 15–64 years. Baseline tariffs are sectoral input and output tariffs in the baseline year.

Source: authors' elaboration on the census, NLF, and WITS data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014; World Bank 2021).

Our main explanatory variables are the district-level time-varying measures of exposure to output and input tariff reduction, which were defined in Section 2. The cross-sectional variation in initial conditions is crucial to have heterogenous responses across districts to nation-wide tariffs reduction.

Our baseline panel specification is therefore the following:

$y_{k,t} = \alpha + \beta_1 Output Tarif f_{k,t} + \beta_2 Input Tarif f_{k,t} + \gamma X_{k,t} + \delta_t + \pi_k + \varepsilon_{k,t}$

where $y_{k,t}$ represent one of our outcomes, namely general and disaggregated levels of employment for district k and at time t, but also unemployment rate and educational attainment. Our main coefficients of interest are β_1 and β_2 , which show the average effect of tariffs reduction on $y_{k,t}$. If β is greater (lower) than zero, this means that a reduction in tariffs is associated with a reduction (increase) in employment (or more in general our variable of interest) $y_{k,t}$.

To tackle the potential omitted variable bias, we add a vector of time-varying controls at the district level, $X_{k,t}$. Controls include the share of urban population within each district, to capture the effect

of urbanization movements, the share of illiterate people, to capture human capital accumulation, and the average age of the population. We also add year fixed effects (δ_t), to control for common shocks affecting all districts and, especially, district-level fixed effects (π_k), to control for potential time-invariant district-specific unobserved factors.¹⁷ Standard errors are clustered at the district level, to account for potential autocorrelation within districts over time. Although changes in the two tariff variables are correlated, this should not affect our estimates. Their partial correlation, once all covariates are controlled for, amounts to around 0.15. To be sure that our identification assumption is robust, we test our results through alternative specifications, in first difference and using the volume of imports as the main explanatory variable, whose results are reported in Section 4.2.

4 Results

4.1 Main results

Table 3 shows the main results of our baseline specification. In Column 1 we regress the district levels of employment against the district exposure to a reduction in output tariffs, plus the abovementioned set of covariates, district and year fixed effects; what we observe is a detrimental impact of the reduction in output tariffs on the general level of employment in the Ethiopian districts. In Column 2, we isolate the effects of an exposure to a reduction in input tariffs, which turns out to be negligible. Note, however, that if we control for both effects simultaneously, as in Column 3, we can see that the negative effect of a reduction in output tariffs remains (and slightly increases in magnitude), whereas the coefficient of the input tariffs becomes negative and significant, suggesting that in the specification of Column 2 there could be some omitted variable bias. Overall, for the average district, a 1-point reduction in output tariffs is associated with a 2.1 percentage point decrease in employment, but at the same time is also associated with a 0.2 percentage point increase in employment, due to an equivalent reduction on input prices. However, if we consider that, on average, the drop in output tariffs (1.097) is about three times larger than the one of input tariffs (0.389), there will be a net reduction of employment due to trade liberalization. Therefore, the first takeaway message is that in Ethiopia trade liberalization, proxied by a reduction in output tariffs, has not resulted in pro-competitive effects but rather it seems to have had disruptive effects on employment; at the same time, the possibility of using cheaper inputs had (even though smaller) positive effects on district employment.

The robustness of this first finding is tested in two ways: first, switching specification to a firstdifference approach [used in a similar context, for instance, by Kis-Katos et al. (2018) and Lepelle and Edwards (2020)]; second, changing the definition of the explanatory variable, from (output) tariffs to the volume of imports. We run the same analysis of the baseline, but employing a firstdifference specification, where changes in employment rates are related to changes in tariff measures, and report the results in Column 1 of Appendix Table A2. As in the baseline, we include the same set of control variables. The direction and significance of our estimates do not change, despite the fact that the sample is smaller. In Column 2 of Appendix Table A2 we use import

¹⁷ To control for district-specific unobserved factors, we could use a fixed-effect or random-effect specification. However, the random-effect model requires that the omitted variables are uncorrelated with the other covariates, which seem unrealistic in our case. In any case, to check the validity of our specification, we perform the postestimation Hausman test.

values instead of (output) tariffs.¹⁸ The coefficient is negative, and therefore the result is consistent with the previous, showing that an increase in imports leads to a decrease in general employment.

Variables	OLS	OLS	OLS
	Employed	Employed	Employed
Output tariffs	0.022***	_	0.027***
	(0.003)		(0.004)
Input tariffs	—	0.001	-0.003***
		(0.001)	(0.001)
Urban	-0.115***	-0.126***	-0.115***
	(0.035)	(0.036)	(0.035)
Age	0.003	0.003	0.004
	(0.003)	(0.003)	(0.003)
Illiterate	0.134***	0.148***	0.118***
	(0.041)	(0.042)	(0.042)
Constant	0.457***	0.560***	0.488***
	(0.093)	(0.095)	(0.092)
Mean of dependent variable	0.78	0.78	0.78
Observations	1,285	1,285	1,285
<i>R</i> -squared	0.429	0.405	0.433
Number of districts	388	388	388
District fixed effects	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes

Table 3: Baseline results

Note: OLS, ordinary least square; SE, standard error. p<0.1, p<0.05, p<0.05, p<0.01. SE (in parentheses) are clustered at the district level. Note that the employment rate is calculated by dividing the number of people aged 15–64 years in employment by the total population of the same age group.

Source: authors' elaboration on the census, NLF, and WITS data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014; World Bank 2021).

4.2 Differences across gender

Next, we check whether there are differences in terms of employment levels to liberalization across the gender of workers. Results are reported in Table 4. The magnitude of the output tariff coefficient is almost three times larger for women (Column 2) than for men (Column 1); a fact in line with Figure 5, where we showed that more female-intensive sectors experienced a faster reduction in output tariffs. Input tariffs, instead, seem to have a significant and positive effect only on female employment. To test whether input and output tariffs are statistically significant across gender, we perform a Wald test and find that there is a significant differential impact of tariff reductions between female and male workers. A 1-point reduction in output tariffs is associated with a 1 percentage point reduction in male employment but also with a 3 percentage point

Import index_{k,t} =
$$\sum_{s=1}^{S} \frac{Empl_{s,k,1994}}{Empl_{k,1994}} \times \ln (Imports)_{s,t}$$

¹⁸ For this test, following Autor et al. (2016), we assess the impact of trade liberalization using changes in the volume of imports instead of changes in tariffs. Therefore, we construct a measure similar to tariff measure, but using real imports (in US\$), as illustrated below:

where Ethiopian industry-specific imports from all over the world (in logs) are weighted using employment shares for each district.

reduction in female employment. At the same time, the same reduction in input tariffs has no effect on male employment but increases female employment by a 0.5 percentage point. Overall, it seems that trade liberalization, besides having a negative impact on the general employment level, also widens the gender employment gap, penalizing women more than men; the negative effect of output tariffs seems to be not totally compensated by the positive effect that input tariff liberalization has on women. Also in this case, results are robust to the specification in first differences (Columns 1 and 2 in Appendix Table A3).

Variables	OLS	OLS
	Employed male	Employed female
Output tariffs	0.012***	0.042***
	(0.003)	(0.005)
Input tariffs	0.001	-0.007***
	(0.001)	(0.002)
Constant	0.625***	0.325**
	(0.083)	(0.136)
Mean of dependent variable	0.87	0.69
Observations	1,285	1,285
<i>R</i> -squared	0.380	0.366
Controls	Yes	Yes
Number of districts	388	388
District fixed effects	Yes	Yes
Year fixed effects	Yes	Yes
Cluster SE	Yes	Yes

Table 4: Differences across gender

Note: OLS, ordinary least square; SE, standard error. *p<0.1, **p<0.05, ***p<0.01. SE (in parentheses) are clustered at the district level. Note that the employment rate of men (women) is calculated by dividing the number of men (women) aged 15–64 years in employment by the total male (female) population of the same age group. Controls include the share of urban population within each district, the share of illiterate, and the average age of the population.

Source: authors' elaboration on the census, NLF, and WITS data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014; World Bank 2021).

5 Mechanisms and extensions

Our results point to a drop in the levels of employment following episodes of trade liberalization. We also show that this drop is apparently more consistent for women than for men, even though there is also a small but significant push of female employment because of the reduction in input tariffs. In this section, we try to unpack some of these findings by showing potential patterns of reallocation of workers across sectors and types of employment. We do this by running our analysis first on total employment and then across genders. Finally, and again with a gender perspective in mind, we check to what extent reallocation due to trade liberalization has come along with more or less segregation across economic activities by women and we explore some extensions to our baseline results, looking at unemployment, education, and reallocation within services.

5.1 Reallocation across economic activities

We ask whether labour reallocation dynamics across sectors a consequence of trade liberalization is. We differentiate between the three macro-sectors: agriculture, manufacturing, and services and explore the relationship between the reduction in the output and input tariffs and employment shares in the three sectors.¹⁹

Figure 7 plots the coefficients of interest and the full set of results is reported in Appendix Table A4. There is a significant job disruption effect of tariffs in the agricultural sector. This dynamic is mainly driven by the input tariff reduction, whereas the coefficient for output tariffs is not significantly different from zero (green bars, 1 and 4). In the manufacturing sector instead, a reduction in output tariffs is associated with an increase in the general employment level, whereas input tariffs do not seem to play any role in the overall story. The negative coefficients (3 and 6) suggest instead that tariff liberalization induced significant worker reallocation towards the tertiary sector [a result in line, for instance, with what happened during the Brazilian trade reforms in the 1990s, as shown by Menezes-Filho and Muendler (2011) and Gaddis and Pieters (2017)]. Again, this is mainly driven by lower input tariffs. However, looking at the means of the dependent variables, the employment gains in the tertiary sector cannot make up for the overall losses in the primary sector, which is in line with the generalized trade-induced job losses found in the previous sections.



Figure 7: Reallocation across sectors

¹⁹ Note that here the shares of workers in agriculture, manufacturing, and services are expressed as the number of employed persons in each sector over the total number of employed people in working age (15–64 years) in each district. In this way, by keeping the denominator fixed and equal to the total employment level, we can see the reallocation of the employed across sectors due to tariff liberalization.

We perform the same analysis disaggregated by gender and the results are reported in Appendix Table A5. As we can see, the dynamics from the primary to the tertiary sector seems to be more pronounced for female employment than for male employment. Next, we try to disentangle the effect of the reduction in tariffs by creating two different indices, one covering only tariffs of the agricultural sector and the other confined to the manufacturing sector. We then replicate the same analysis. Results almost identically replicate those obtained when using tariffs of agricultural sector (Appendix Table A6). This is not surprising given the importance of the primary sector in terms of employment in most areas of the country. On the other hand, we observe limited implications due to the drop in manufacturing tariffs (Appendix Table A7), whose effects are mostly found on within-sector changes in employment. Specifically, an increase in competition (proxied by a reduction in output tariffs) contributes to a reduction of (female) employment in the manufacturing sector, whereas a decrease in input tariffs (cheaper inputs) raises it.

Therefore, an additional takeaway message from our study is that trade liberalization in Ethiopia has contributed to some forms of structural transformation, following a pattern of structural transformation without industrialization that seems nowadays typical of laggard economies in the continent (Rodrik 2016). We show that for Ethiopia this phenomenon has involved mostly women. However, similar to Lepelle and Edwards (2020), the decline in employment in the liberalized sector (agriculture in our case, manufacturing in theirs) has not been compensated by an even increase in services.

5.2 Sectoral segregation

The fact that female workers leave the primary sector following trade liberalization can be correlated with what has been found by other studies (discussed in Section 1, in particular Gaddis and Pieters 2017). Indeed, women might be more vulnerable to an increase in imports, given that they are more likely to be engaged in less-productive substance farming (O'Sullivan et al. 2014). Moreover, the country has an estimated 29.5 per cent gender gap in agricultural productivity because female managers use fewer inputs than their male counterparts (WB and WTO 2020). In recent years, the country has experienced a higher adoption of modern inputs in agriculture (together with an increase in urbanization, in the road infrastructure, and in the communication infrastructure; see Minten et al. 2014), which might have contributed to push the least productive workers, among which women are the majority, out of the primary sector. Unfortunately, as we do not have information on productivity nor on the nature of the specific inputs used in each and every productive activity, we cannot test this mechanism.

As we find that output tariffs deteriorate women's employment, changes in discrimination are unlikely to be driving our results. Another channel through which trade liberalization might affect female and male workers differently is sectoral segregation. If particularly stringent social norms are in place, it may be that the structural transformation induced by trade liberalization would take place with differences between gender, resulting in an increase in sectoral segregation between men and women. Naturally, as argued by Wang et al. (2020), the opposite (i.e. a de-segregation) might happen too. Indeed, it is also plausible that an increasing exposure to international markets can shift social norms and lower the preference for a segregated labour market. To quantitatively assess the role played by sectoral segregation, we construct the index of dissimilarity, first proposed by Duncan and Duncan (1955) and then adopted in some related works (Borrowman and Klasen 2020; Kis-Katos et al. 2018; Wang et al. 2020):

$$ID_{k,t} = \frac{1}{2} \sum_{s=1}^{S} \left| \left(\frac{M_{skt}}{M_{kt}} - \frac{F_{skt}}{F_{kt}} \right) \right|$$

where $\frac{M_{skt}}{M_{kt}}$ and $\frac{F_{skt}}{F_{kt}}$, respectively, measure the share of sector s for total male employment and female employment at time t in district k. For the way it is constructed, an increase (reduction) in the index value, that goes from 0 to 1, means that sectors become more (less) segregated by gender. As shown in Table 5, a reduction in input tariffs is correlated with an increase in sectoral segregation, meaning that the average distance in terms of sectoral presence between male and female workers has *increased* following trade liberalization, a fact that is probably linked to the gendered sectoral reallocation observed in the previous section. Our results contrast with those of Kis-Katos et al. (2018) for Indonesia and Wang et al. (2020) for China, who find that lower input tariffs went hand-in-hand with a reduction in sectoral segregation. However, they also find that, overall, trade liberalization contributed to the feminization of the economy, increasing female participation in the labour market, a result that does not seem to apply to the Ethiopian context (possibly because of the low relevance of the manufacturing sector in Ethiopia compared with China and Indonesia). Therefore, what our results seem to suggest is that the sectoral reallocation that has taken place following trade liberalization pushed women out of the agricultural sector towards the services sector. However, structural transformation has increased sectoral segregation, increasing the gap in terms of female and male presence in each sector.

Variables	(1)			
	Segregation index			
Output tariffs	-0.004			
	(0.005)			
Input tariffs	-0.003*			
	(0.002)			
Constant	0.345**			
	(0.169)			
Observations	1,285			
Number of districts	388			
R-squared	0.498			
Year fixed effects	Yes			
Controls	Yes			
District fixed effects	Yes			
Cluster SE	Yes			

Table 5: Sectoral segregation

Note: SE, standard error. *p<0.1, **p<0.05, ***p<0.01. SE (in parentheses) are clustered at the district level. Controls include the share of urban population within each district, the share of illiterate, and the average age of the population.

Source: authors' elaboration on the census, NLF, and WITS data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014; World Bank 2021).

5.3 Extensions: unemployment, years of education, and services decomposition

We conclude our analysis by looking at some dimensions potentially related to employment. On the ground of our main finding showing that trade liberalization brought a net reduction in employment, we investigate some of the potential strategies undertaken by individuals in response to such an occurrence. First, we look at unemployment levels.²⁰ Although reported unemployment in Ethiopia is extremely low throughout the period covered by our study (around 2 per cent for all four waves), we can see how an increase in market competition due to lower output tariffs is associated with a small but significant increase in the general unemployment level (Appendix Table A8, Column 1), a result that turns out to be mainly driven by male unemployment.

Second, we look at the average years of education within districts over time. We find that a reduction of output tariffs has no effect on a district's average education level, whereas more exposed districts experienced a reduction in women's educational uptakes as a consequence of a reduction in input prices (Appendix Table A9).²¹ This result, read together with our finding that input tariff reduction led to a small but significant increase of female employment, might tell us a peculiar story: a reduction in input prices, through the adoption of cheaper labour-saving inputs, might have freed up (female) workers from the agricultural sector in more exposed districts and reallocated them to the services sector. However, this has not provided incentives for women to invest more in education. A potential explanation for that might be that the relocation to the services sector has predominantly taken place towards low-skilled and low-paying sectors.²²

To test this hypothesis, we analyse which of the service sectors grew the most following trade liberalization.²³ In line with what we expected, the major shifts following a decrease in input tariffs were for female workers, and towards those sectors that can be reasonably assumed to be low-skilled (Appendix Tables A10 and A11). This includes, for instance, 'Community services' (division O), and 'Private household activities' (division P).²⁴ Reductions in input tariffs are associated also with an increase in male employment within those sectors, even though of a smaller size. At the same time, we see no significant effects of trade liberalization for men or women in high-skilled market services (e.g. 'Finance and real estate', divisions J–K). Finally, we find some evidence of an increase in the employment in 'Public administration services' (division L) for both men and women following a reduction in output tariffs.

 $^{^{20}}$ We define as unemployed people aged 15–64 years who are not in employment but carry out activities to seek employment.

²¹ Also, Edmonds et al. (2010) find that in rural India schooling improvements were attenuated in districts with employment concentrated in industries losing tariff protection, and that was particularly true for girls, even though they point to the poverty–schooling costs relationship as a potential mechanism. We of course do not exclude it, but we have insufficient data to test it.

²² This role played by trade-induced structural change (or, as in our case, by sectoral reallocation), in determining returns to education that, in turn, determine educational attainment, has been investigated for instance also by Foster and Rosenzweig (1996).

²³ We categorize services following the ISIC 3.1 classification that divides sectors into divisions. The divisions we use are: E–F (Electricity and construction), G (Wholesale and retail), H (Hotels and restaurants), I (Transports), J–K (Finance and real estate), L (Public admin.), M–N (Health and education), O (Community services), and P (Private household activities).

²⁴ The subsectors included in divisions O and P are: Sewage and refuse disposal, Activities of membership organizations, Recreational, cultural and sporting activities, and Other service activities in division O, and Activities of private households as employers of domestic staff, Undifferentiated goods-producing activities of private households for own use, and Undifferentiated service-producing activities of private households for own use in division P.

6 Conclusions

This paper shows that districts in Ethiopia that were more exposed to import tariff reductions experienced a stronger reduction in employment levels compared with districts less exposed to international competition. This result is robust to several specifications. The direction of the results is consistent with existing literature (e.g., Dix-Carneiro and Kovak 2017; Topalova 2007). From a gender perspective, we find that tariff reforms have a more detrimental effect on female employment than on male employment. We show also that trade liberalization seems to affect the distribution of employment out of the agricultural sector and towards services, again with more marked effects for female workers. Women are in fact those that exit faster from the primary sector and enter into the services sector following a reduction in input tariffs, although this process was not accompanied by a reduction of sectoral segregation affecting women's employment. Moreover, we see that lower input tariffs, although facilitating the migration to the services sector, discourage women from increasing their educational attainment and push them to move into low value-added sectors such as private household activities.

References

- Amiti, M., and J. Konings (2007). 'Trade Liberalization, Intermediate Inputs, and Productivity: Evidence from Indonesia'. *American Economic Review*, 97(5): 1611–38. https://doi.org/10.1257/aer.97.5.1611
- Attanasio, O., P.K. Goldberg, and N. Pavcnik (2004). "Trade Reforms and Wage Inequality in Colombia'. Journal of Development Economics, 74(2): 331–66. https://doi.org/10.1016/j.jdeveco.2003.07.001
- Autor, B.D.H., D. Dorn, and G.H. Hanson (2016). 'The China Syndrome: Local Labor Market Effects of Import Competition in the United States'. The American Economic Review, 103(6): 2121–68. https://doi.org/10.1257/aer.103.6.2121
- Becker, G.S. (1957). The Economics of Discrimination. Chicago, IL: University of Chicago Press.
- Ben Yahmed, S., and P. Bombarda (2020). 'Gender, Informal Employment and Trade Liberalization in Mexico'. World Bank Economic Review, 34(2): 259–83. https://doi.org/10.1093/wber/lhy020
- Bigsten, A., M. Gebreeyesus, and M. Söderbom (2016). 'Tariffs and Firm Performance in Ethiopia'. *Journal of Development Studies*, 52(7): 986–1001. https://doi.org/10.1080/00220388.2016.1139691
- Black, S., and E. Brainerd (2004). 'Import Equality? The Impact of Globalization on Gender Discrimination'. *ILR Review*, 57(4): 540–59. https://doi.org/10.1177/001979390405700404
- Borrowman, M., and S. Klasen (2020). 'Drivers of Gendered Sectoral and Occupational Segregation in Developing Countries'. *Feminist Economics*, 26(2): 62–94. https://doi.org/10.1080/13545701.2019.1649708
- Cirera, X., D. Willenbockel, and R.W.D. Lakshman (2014). 'Evidence on the Impact of Tariff Reductions on Employment in Developing Countries: A Systematic Review'. *Journal of Economic Surveys*, 28(3): 449– 71. https://doi.org/10.1111/joes.12029
- CSA (1994). Ethiopia: Population and Housing Census 1994. Addis Ababa: Central Statistics Agency (CSA). Available at: https://www.statsethiopia.gov.et/census-1994/ (accessed February 2021).
- CSA (2004). 'Analytical Report of the National Labour Force Survey'. *Statistical Bulletin of the Central Statistical Authority*. Addis Ababa: Central Statistics Agency (CSA). Available at: https://catalog.ihsn.org//catalog/156/download/29150 (accessed February 2021).
- CSA (2006). 'Analytical Report of the National Labour Force Survey'. *Statistical Bulletin of the Central Statistical Authority*. Addis Ababa: Central Statistics Agency (CSA). Available at: https://catalog.ihsn.org//catalog/3753/download/52232 (accessed February 2021).

- CSA (2014). 'Analytical Report of the National Labour Force Survey'. *Statistical Bulletin of the Central Statistical Authority*. Addis Ababa: Central Statistics Agency (CSA). Available at: https://www.ilo.org/public/libdoc//igo/P/392984/392984(2013)480.pdf (accessed February 2021).
- Diao, X., M. Ellis, M. McMillan, and D. Rodrik (2021). 'Africa's Manufacturing Puzzle: Evidence from Tanzanian and Ethiopian Firms'. NBER Working Paper Series 28344. Cambridge, MA: National Bureau of Economic Research (NBER). https://doi.org/10.3386/w28344
- Dix-Carneiro, B.R., and B.K. Kovak (2017). 'Trade Liberalization and Regional Dynamics'. *The American Economic Review*, 107(10): 2908–46. https://doi.org/10.1257/aer.20161214
- Duncan, D., and B. Duncan (1955). 'A Methodological Analysis of Segregation Indexes'. *American Sociological* Review, 20(2): 210–17. https://doi.org/10.2307/2088328
- Easterly, W. (2003). 'IMF and World Bank Structural Adjustment Programs and Poverty'. In M.P. Dooley and J.A. Frankel (eds), *Managing Currency Crises in Emerging Markets* (pp. 361–92). Chicago, IL: University of Chicago Press. https://doi.org/10.7208/chicago/9780226155425.003.0012
- Ederington, J., J. Minier, and K.R. Troske (2009). 'Where the Girls Are: Trade and Labor Market Segregation in Colombia'. IZA Discussion Paper 4131. Bonn: Institute of Labor Economics (IZA). Available at: https://www.iza.org/publications/dp/4131/where-the-girls-are-trade-and-labormarket-segregation-in-colombia (accessed February 2021).
- Edmonds, E.V., N. Pavcnik, and P. Topalova (2010). 'Trade Adjustment and Human Capital Investments: Evidence from Indian Tariff Reform'. *American Economic Journal: Applied Economics*, 2(4): 42–75. https://doi.org/10.1257/app.2.4.42
- Edwards, L. (2004). 'A Firm Level Analysis of Trade, Technology and Employment in South Africa''. *Journal of International Development*, 16(1): 45–61. https://doi.org/10.1002/jid.1062
- Erten, B., J. Leight, and F. Tregenna (2019). 'Trade Liberalization and Local Labor Market Adjustment in South Africa'. *Journal of International Economics*, 118: 448–67. https://doi.org/10.1016/j.jinteco.2019.02.006
- Essers, D., K. Megersa, and M. Sanfilippo (2021). "The Productivity Gaps of Female-Owned Firms: Evidence from Ethiopian Census Data'. *Economic Development and Cultural Change*, 69(2): 645–83. https://doi.org/10.1086/703101
- Fajgelbaum, P., and S. Redding (2014). 'External Integration, Structural Transformation and Economics Development: Evidence from Argentina 1870–1914'. NBER Working Paper Series 20217. Cambridge, MA: National Bureau of Economic Research (NBER). https://doi.org/10.1086/703101
- Feliciano, Z.M. (2001). 'Workers and Trade Liberalization: The Impact of Trade Reforms in Mexico on Wages and Employment'. *Industrial and Labor Relations Review*, 55(1): 95–115. https://doi.org/10.1177/001979390105500106
- Fiorini, M., and M. Sanfilippo (2019). 'Roads and Jobs in Ethiopia'. WIDER Working Paper 116/2019. Helsinki: UNU-WIDER. https://doi.org/10.35188/UNU-WIDER/2019/752-1
- Foster, A., and M. Rosenzweig (1996). 'Technical Change and Human-Capital Returns and Investments: Evidence from the Green Revolution'. *The American Economic Review*, 86(4): 931–53.
- Gaddis, I., and J. Pieters (2017). "Trade Liberalization and Female Labor Force Participation: Evidence from Brazil'. *Journal of Human Resources*, 52(2): 457–90. https://doi.org/10.3368/jhr.52.2.1014-6690R1
- Gebreeyesus, M. (2016). 'Industrial Policy and Development in Ethiopia'. *Manufacturing Transformation*, 6: 27–49. https://doi.org/10.1093/acprof:oso/9780198776987.003.0002
- Goldberg, P.K., A.K. Khandelwal, N. Pavcnik, and P. Topalova (2010). 'Imported Intermediate Inputs and Domestic Product Growth: Evidence from India'. *Quarterly Journal of Economics*, 125(4): 1727–67. https://doi.org/10.1162/qjec.2010.125.4.1727

- Goldberg, P.K., and N. Pavcnik (2007a). 'The Effects of the Colombian Trade Liberalization on Urban Poverty'. In A. Harrison (ed.), *Globalization and Poverty* (pp. 241–90). Cambridge, MA: National Bureau of Economic Research.
- Goldberg, P.K., and N. Pavcnik (2007b). 'Distributional Effects of Globalization in Developing Countries'. Journal of Economic Literature, 45(1): 39–82. https://doi.org/10.1257/jel.45.1.39
- Goldin C (1995). 'The U-Shaped Female Labor Force Function in Economic Development and Economic History'. In T.P. Schultz (ed.), *Investment in Women's Human Capital* (pp. 61–90). Chicago, IL: University of Chicago Press. https://doi.org/10.3386/w4707
- Grossman, G.M., and E. Helpman (1994). 'Endogenous Innovation in the Theory of Growth'. *Journal of Economic Perspectives*, 8(1): 23–44. https://doi.org/10.1257/jep.8.1.23
- Hoekman, B., and L.A. Winters (2005). 'Trade and Employment: Stylized Facts and Research Findings'. World Bank Policy Research Working Paper 3676. https://doi.org/10.2139/ssrn.780467
- Jones, C., O. Morrissey, and D. Nelson (2011). 'Did the World Bank Drive Tariff Reforms in Eastern Africa?'. World Development, 39(3): 324-35. https://doi.org/10.1016/j.worlddev.2010.08.003
- Juhn, C., G. Ujhelyi, and C. Villegas-Sanchez (2014). 'Men, Women, and Machines: How Trade Impacts Gender Inequality'. *Journal of Development Economics*, 106: 179–93. https://doi.org/10.1016/j.jdeveco.2013.09.009
- Kis-Katos, K., J. Pieters, and R. Sparrow (2018). 'Globalization and Social Change: Gender-Specific Effects of Trade Liberalization in Indonesia'. *IMF Economic Review*, 66(4): 763–93. https://doi.org/10.1057/s41308-018-0065-5
- Kis-Katos, K., and R. Sparrow (2011). 'Child Labor and Trade Liberalization in Indonesia'. *Journal of Human Resources*, 46(4): 722–49. https://doi.org/10.3368/jhr.46.4.722
- Kis-Katos, K., and R. Sparrow (2015). Poverty, Labor Markets and Trade Liberalization in Indonesia'. Journal of Development Economics, 117: 94–106. https://doi.org/10.1016/j.jdeveco.2015.07.005
- Kovak, B.K. (2013). 'Regional Effects of Trade Reform: What Is the Correct Measure of Liberalization?'. *American Economic Review*, 103(5): 1960–76. https://doi.org/10.1257/aer.103.5.1960
- Kyander, E. (2020). 'Trade Liberalisation and Female Employment in Manufacturing: Evidence from India.' CSAE Working Paper WPS/2020-22. Oxford: Centre for the Study of African Economies (CSAE), University of Oxford.
- Lepelle, R., and L. Edwards (2020). 'Gendered Effects of Tariff Liberalisation on the Sectoral Structure of Employment in South Africa'. PRISM Working Paper Series 2020-4. Cape Town: Policy Research on International Services and Manufacturing (PRISM). Available at: https://webcms.uct.ac.za/sites/default/files/image_tool/images/524/Papers/PRISM%20Working %20Paper%202020-4%20-%20Refilwe%20%26%20Lawrence.pdf (accessed February 2021).
- Meschi, E., and M. Vivarelli (2009). 'Trade and Income Inequality in Developing Countries'. World Development, 37(2): 287–302. https://doi.org/10.1016/j.worlddev.2008.06.002
- Milner, C., and P. Wright (1998). 'Modelling Labour Market Adjustment to Trade Liberalisation in an Industrialising Economy'. *The Economic Journal*, 108(447): 509–28. https://doi.org/10.1111/1468-0297.00301
- Minnesota Population Center (2015). Integrated Public Use Microdata Series, International Version 6.4 [Machine-readable database]. Minneapolis: University of Minnesota. Available at: https://international.ipums.org/international/ (accessed February 2021).
- Minten, B., D. Stifel, and S. Tamru (2014). 'Structural Transformation of Cereal Markets in Ethiopia'. *Journal of Development Studies*, 50(5): 611–29. https://doi.org/10.1080/00220388.2014.887686
- O'Sullivan, M., A. Rao, R. Banerjee, K. Gulati, M. Vinez (2014). Levelling the Field: Improving Opportunities for Women Farmers in Africa. Washington, DC: World Bank Group. Available at: http://documents.worldbank.org/curated/en/579161468007198488/Levelling-the-field-improvingopportunities-for-women-farmers-in-Africa (accessed February 2021).

- Pieters, J. (2018). 'Trade Liberalization and Gender Inequality: Can Free-Trade Policies Help to Reduce Gender Inequalities in Employment and Wages?'. IZA World of Labor, article 114. https://doi.org/10.15185/izawol.114
- Revenga, A. (1997). 'Employment and Wage Effects of Trade Liberalization: The Case of Mexican Manufacturing'. *Journal of Labor Economics*, 15(S3). https://doi.org/10.1086/209875
- Rodrik, D. (2016). 'Premature Deindustrialization'. Journal of Economic Growth, 21(1): 1-33. https://doi.org/10.1007/s10887-015-9122-3
- Schultz, T.P. (2007). 'Does the Liberalization of Trade Advance Gender Equality in Schooling and Health?' In E. Zedillo (ed.), *The Future of Globalization: Explorations in Light of Recent Turbulence* (pp. 178–208). London: Routledge. https://doi.org/10.4324/9780203946527
- Sutton, J., and N. Kellow (2010). An Enterprise Map of Ethiopia. London: International Growth Centre. Available at: https://www.theigc.org/project/the-enterprise-map-series/ (accessed February 2021).
- Tebekew, T., A. Amoge, B. Teferra, Z. Seyoum, M. Amha, H. Beyene, E. Fisseha, E. Tsehaye, H.A. Ahmed, S. Robinson, D. Willenbockel, and P.A. Dorosh (2009). *Ethiopia: Input Output Table and Social Accounting Matrix*. Addis Ababa and Washington, DC: Ethiopian Development Research Institute (EDRI) and International Food Policy Research Institute (IFPRI). Available at: http://ebrary.ifpri.org/utils/getfile/collection/p15738coll2/id/128886/filename/129097.pdf (accessed February 2021).
- Topalova, P. (2007). 'Trade Liberalization, Poverty and Inequality: Evidence from Indian Districts'. In A. Harrison (ed.), *Globalization and Poverty* (pp. 291–336). Chicago, IL: University of Chicago Press. https://doi.org/10.7208/chicago/9780226318004.003.0008
- Topalova, P. (2010). 'Factor Immobility and Regional Impacts of Trade Liberalization: Evidence on Poverty from India'. *American Economic Journal: Applied Economics*, 2(4): 1–41. https://doi.org/10.1257/app.2.4.1
- United Nations (1990). International Standard Industrial Classification of All Economic Activities (ISIC) Revision 3. Available at: https://unstats.un.org/unsd/classifications/Family/Detail/2 (accessed February 2021).
- Wang, F., K. Kis-Katos, and M. Zhou (2020). "Trade Liberalization and the Gender Employment Gap in China'. GLO Discussion Paper 638 (Issue July). Essen: Global Labor Organization (GLO). Available at: https://www.econstor.eu/bitstream/10419/223048/1/GLO-DP-0638.pdf (accessed February 2021).
- WB and WTO (2020). Women and Trade. The Role of Trade in Promoting Gender Equality. Washington, DC: International Bank for Reconstruction and Development/World Bank (WB) Group and the World Trade Organization (WTO). Available at: https://www.wto.org/english/res_e/publications_e/women_trade_pub2807_e.htm (accessed February 2021).
- Winters, L.A., and M. Di Ubaldo (2020). International Trade Regulation and Job Creation'. *IZA World of Labor*, February, 1–10: article 75. https://doi.org/10.15185/izawol.75.v2
- World Bank (2021). WITS Database. Available at: https://wits.worldbank.org/tariff/trains/country-byhs6product.aspx?lang=en (accessed February 2021).

Appendix

Year	Mean	SD	Kurtosis	Skewness
1995	24.85	17.02	4.04	0.92
1999	19.88	12.44	2.8	0.53
2005	16.14	10.06	2.11	0.34
2013	15.23	9.623	2.3	0.44

Note: SD, standard deviation.

Source: authors' elaboration on WITS data and IFPRI SAM (World Bank 2021; Tebekew et al. 2009).

Table A2: FD and imports—robustness checks

Variables	FD	OLS
	Δ Employed	w/ Imports
Imports		-0.025**
		(0.010)
Δ Output	0.043***	—
	(0.007)	
Δ Input index	-0.004**	—
	(0.002)	
Δ Urban	-0.151***	-0.109***
	(0.040)	(0.037)
Δ Illiterate	0.055	0.164***
	(0.049)	(0.042)
ΔAge	0.005	0.003
	(0.004)	(0.003)
Constant	-0.071***	0.662***
	(0.024)	(0.100)
Observations	750	1,285
Number of districts	291	388
Year fixed effects	Yes	Yes
District fixed effects	No	Yes
Cluster SE	Yes	Yes

Note: FD, first difference; OLS, ordinary least square; SE, standard error; *p<0.1, **p<0.05, ***p<0.01. SE (in parentheses) are clustered at the district level.

Table A3: First difference across gender

Variables	FD	FD
	Δ Employed male	Δ Employed female
Output tariffs	0.033***	0.053***
	(0.006)	(0.010)
Input tariffs	-0.001	-0.006**
	(0.001)	(0.003)
Urban	-0.092**	-0.201***
	(0.040)	(0.061)
Age	0.123***	-0.004
	(0.043)	(0.070)
Illiterate	0.002	0.010*
	(0.003)	(0.006)
Constant	-0.045**	-0.097***
	(0.020)	(0.035)
Observations	750	750
Number of districts	291	291
District fixed effects	No	No
Year fixed effects	Yes	Yes
Cluster SE	Yes	Yes

Note: p<0.1, p<0.05, p<0.05, p<0.01. SE (in parentheses) are clustered at the district level. The employment rate of men (women) is calculated by dividing the number of men (women) aged 15–64 years in employment by the total male (female) population of the same age group.

Source: authors' elaboration on the census, NLF, and WITS data (Minnesota Population Center 2015; CSA 1994, 2004, 2006, 2014; World Bank 2021).

Table A4: Reallocation across sectors—complete table

Variables	Agriculture	Manufacturing	Services
Output tariffs	0.005	-0.004**	-0.001
	(0.004)	(0.002)	(0.003)
Input tariffs	0.005***	0.001	-0.006***
	(0.001)	(0.001)	(0.001)
Urban	-0.494***	0.057***	0.441***
	(0.040)	(0.022)	(0.034)
Age	0.001	-0.000	-0.001
	(0.003)	(0.002)	(0.003)
Illiterate	0.340***	-0.093***	-0.229***
	(0.052)	(0.022)	(0.043)
Constant	0.486***	0.092	0.403***
	(0.115)	(0.056)	(0.092)
Mean of dependent variable	0.79	0.05	0.14
Observations	1,285	1,285	1,285
R-squared	0.514	0.202	0.499
Number of districts	388	388	388
Year fixed effects	Yes	Yes	Yes
District fixed effects	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes

Note: p<0.1, p<0.05, p<0.05, p<0.01. SE (in parentheses) are clustered at the district level. Shares of workers in agriculture, manufacturing, and services are expressed as the number of employed persons in each sector over the total number of employed people in working age (15–64 years) in each district.

	(1)	(2)	(3)	(4)	(5)	(6)
Variables	Agriculture, male	Agriculture, female	Manufacturing, male	Manufacturing, female	Services, male	Services, female
Output tariffs	0.003	0.009*	-0.001	-0.007***	-0.001	-0.001
	(0.003)	(0.005)	(0.002)	(0.003)	(0.002)	(0.004)
Input tariffs	0.003**	0.007***	-0.000	0.002***	-0.003***	-0.009***
	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.002)
Constant	0.534***	0.383**	0.135**	0.053	0.302***	0.546***
	(0.103)	(0.172)	(0.053)	(0.090)	(0.070)	(0.144)
Mean of dependent variable	0.84	0.71	0.04	0.06	0.10	0.20
Observations	1,285	1,285	1,285	1,285	1,285	1,285
<i>R</i> -squared	0.521	0.405	0.227	0.123	0.536	0.386
Number of districts	388	388	388	388	388	388
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes	Yes	Yes
District fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes	Yes	Yes	Yes

Table A5: Sectoral reallocation across gender (controls not shown)

Note: **p*<0.1, ***p*<0.05, ****p*<0.01. SE (in parentheses) are clustered at the district level. The employment rate of men (women) is calculated by dividing the number of men (women) aged 15–64 years employed in agriculture, manufacturing, and services by the total male (female) employed population of the same age group. Controls include the share of urban population within each district, the share of illiterate, and the average age of the population.

	(1)	(2)	(3)	(4)	(5)	(6)
Variables	Agriculture, male	Agriculture, female	Manufacturing, male	Manufacturing, female	Services, male	Services, female
Output tariff agriculture	0.003	0.006*	-0.001	-0.005***	-0.001	-0.001
	(0.002)	(0.003)	(0.001)	(0.002)	(0.001)	(0.003)
Input tariff agriculture	0.003**	0.007***	0.000	0.002***	-0.002***	-0.009***
	(0.001)	(0.002)	(0.001)	(0.001)	(0.001)	(0.002)
Urban	-0.492***	-0.486***	0.070***	0.045	0.422***	0.452***
	(0.043)	(0.051)	(0.023)	(0.033)	(0.033)	(0.050)
Age	0.001	0.004	-0.001	-0.000	-0.000	-0.004
	(0.003)	(0.005)	(0.002)	(0.003)	(0.002)	(0.004)
Illiterate	0.399***	0.238***	-0.119***	-0.048	-0.261***	-0.171**
	(0.048)	(0.080)	(0.022)	(0.032)	(0.035)	(0.074)
Constant	0.543***	0.399**	0.138**	0.063	0.292***	0.518***
	(0.103)	(0.170)	(0.053)	(0.090)	(0.069)	(0.143)
Mean of dependent variable	0.85	0.72	0.04	0.06	0.10	0.20
Observations	1,285	1,285	1,285	1,285	1,285	1,285
<i>R</i> -squared	0.520	0.405	0.228	0.123	0.535	0.385
Number of districts	388	388	388	388	388	388
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
District fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes	Yes	Yes	Yes

Table A6: Reallocation across sectors—agricultural tariffs

Note: **p*<0.1, ***p*<0.05, ****p*<0.01. SE (in parentheses) are clustered at the district level. Shares of workers in agriculture, manufacturing, and services are expressed as the number of employed persons in each sector over the total number of employed people in working age (15–64 years) in each district.

	(1)	(2)	(3)	(4)	(5)	(6)
Variables	Agriculture, male	Agriculture, female	Manufacturing, male	Manufacturing, female	Services, male	Services, female
Output tariff manufacturing.	0.036	0.023	-0.008	0.021**	-0.027	-0.043
	(0.033)	(0.038)	(0.015)	(0.010)	(0.019)	(0.032)
Input tariff manufacturing	-0.069	0.009	0.039	-0.058**	0.027	0.041
	(0.069)	(0.097)	(0.036)	(0.026)	(0.037)	(0.078)
Urban	-0.503***	-0.478***	0.079***	0.050	0.422***	0.441***
	(0.044)	(0.056)	(0.023)	(0.034)	(0.033)	(0.054)
Illiterate	0.403***	0.294***	-0.122***	-0.052	-0.265***	-0.221***
	(0.048)	(0.078)	(0.022)	(0.032)	(0.034)	(0.071)
Constant	0.660***	0.689***	0.097***	0.063**	0.229***	0.228***
	(0.039)	(0.066)	(0.018)	(0.027)	(0.028)	(0.060)
Mean of dependent variable	0.85	0.72	0.04	0.06	0.10	0.20
Observations	1,240	1,240	1,240	1,240	1,240	1,240
<i>R</i> -squared	0.511	0.390	0.247	0.123	0.516	0.354
Number of districts	370	370	370	370	370	370
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
District fixed effects	Yes	Yes	Yes	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes	Yes	Yes	Yes

Table A7: Reallocation across sectors-manufacturing tariffs

Note: *p<0.1, **p<0.05, ***p<0.01. SE (in parentheses) are clustered at the district level. Shares of workers in agriculture, manufacturing, and services are expressed as the number of employed persons in each sector over the total number of employed people in working age (15–64 years) in each district.

Table A8: Unemployment levels

	OLS	OLS	OLS
Variables	Unemployed	Unemployed male	Unemployed female
Output tariffs	-0.001***	-0.002***	-0.001
	(0.000)	(0.001)	(0.001)
Input tariffs	-0.000	-0.000	0.000
	(0.000)	(0.000)	(0.000)
Urban	0.047***	0.041***	0.054***
	(0.009)	(0.009)	(0.012)
Illiterate	-0.023***	-0.014**	-0.033***
	(0.006)	(0.006)	(0.008)
Age	-0.000	-0.000	-0.000
	(0.000)	(0.000)	(0.000)
Constant	0.036**	0.037**	0.041**
	(0.015)	(0.017)	(0.018)
Mean of dependent variable	0.01	0.01	0.01
Observations	1,285	1,285	1,285
<i>R</i> -squared	0.260	0.181	0.256
Number of districts	388	388	388
District fixed effects	Yes	Yes	Yes
Year fixed effects [4]	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes

Note: *p<0.1, **p<0.05, ***p<0.01. SE (in parentheses) are clustered at the district level. Shares of workers in agriculture, manufacturing, and services are expressed as the number of employed persons in each sector over the total number of employed people in working age (15–64 years) in each district.

Table A9: Effects on years of education

	OLS	OLS	OLS
Variables	Years of school	Years of school, male	Years of school, female
Output tariffs	0.019	0.021	0.018
	(0.011)	(0.014)	(0.015)
Input tariffs	0.003	-0.003	0.007**
	(0.003)	(0.003)	(0.004)
Urban	1.139***	1.306***	1.001***
	(0.160)	(0.204)	(0.164)
Age	-0.012	-0.019	0.006
	(0.011)	(0.013)	(0.014)
Illiterate	-6.018***	-7.085***	-5.020***
	(0.190)	(0.225)	(0.236)
Constant	5.734***	7.306***	3.933***
	(0.350)	(0.482)	(0.406)
Mean of dependent variable	1.68	2.26	1.14
Observations	1,285	1,285	1,285
<i>R</i> -squared	0.914	0.894	0.846
Number of districts	388	388	388
Year fixed effects	Yes	Yes	Yes
District fixed effects	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes

Note: *p<0.1, **p<0.05, ***p<0.01. SE (in parentheses) are clustered at the district level. Shares of workers in agriculture, manufacturing, and services are expressed as the number of employed persons in each sector over the total number of employed people in working age (15–64 years) in each district.

Table A10: Reallocation across services—female workers

	СС	C	U	I	LK	1		0	D
		G		I	J-K	L	IVI—IN	0	۲
Variables	Electricity and	Wholesale	Hotels and	Transports	Finance and	Public	Health and	Community	Private
	construction	and retail	restaurants		real estate	administration	education	services	household
									activities
Output tariffs	-0.000	0.003	-0.001	0.000*	0.000	-0.002***	-0.002	0.002	0.002
	(0.000)	(0.002)	(0.002)	(0.000)	(0.000)	(0.001)	(0.001)	(0.002)	(0.003)
Input tariffs	0.000	0.001	-0.001	0.000	-0.000	0.000	0.000	-0.004***	-0.008***
	(0.000)	(0.001)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.002)
Urban	0.000	0.132***	0.130***	0.007***	0.009	0.044***	0.022	0.104***	0.096***
	(0.000)	(0.025)	(0.029)	(0.002)	(0.008)	(0.013)	(0.013)	(0.025)	(0.029)
Age	-0.000	-0.004*	-0.002	0.000	0.000	0.001	0.001	0.000	-0.002
	(0.000)	(0.002)	(0.002)	(0.000)	(0.000)	(0.001)	(0.001)	(0.002)	(0.003)
Illiterate	0.000	-0.102***	-0.059	-0.004	-0.018***	-0.028**	-0.035***	-0.009	0.078*
	(0.000)	(0.033)	(0.037)	(0.004)	(0.007)	(0.011)	(0.011)	(0.027)	(0.043)
Constant	-0.000	0.180**	0.170***	0.002	0.011	-0.005	-0.011	0.100	0.194*
	(0.000)	(0.070)	(0.060)	(0.004)	(0.007)	(0.025)	(0.036)	(0.076)	(0.104)
Mean of dependent variable	0.000	0.07	0.052	0.000	0.002	0.007	0.012	0.030	0.047
Observations	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285
<i>R</i> -squared	0.012	0.202	0.245	0.092	0.137	0.091	0.049	0.167	0.366
Number of districts	388	388	388	388	388	388	388	388	388
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
District fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Note: **p*<0.1, ***p*<0.05, ****p*<0.01. SE (in parentheses) are clustered at the district level. Sectors have been divided following the ISIC 3.1 classification's division in sections. Shares of female workers in the service sectors are expressed as the number of employed persons in each sector over the total number of employed women in working age (15–64 years) in each district.

T I I A A A				
12010 0111	Regulación	arrace	eanvicae_	_maia workare
	Reallocation	201033	30111003	

	E–F	G	Н	I	J–K	L	M–N	0	Р
Variables	Electricity and construction	Wholesale and retail	Hotels and restaurants	Transports	Finance and real estate	Public administration	Health and education	Community services	Private household activities
Output tariffs	0.000	0.001	0.000	0.000	0.000	-0.002**	-0.002	0.001	0.000
	(0.000)	(0.001)	(0.000)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.000)
Input tariffs	-0.000	-0.000	-0.000	-0.000*	-0.000**	-0.000*	-0.000*	-0.001**	-0.001*
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Urban	-0.001	0.150***	0.023***	0.025***	0.019**	0.112***	0.039***	0.052***	0.005
	(0.001)	(0.026)	(0.004)	(0.007)	(0.008)	(0.016)	(0.013)	(0.011)	(0.005)
Age	-0.000	-0.001	-0.000	0.000	0.000	0.000	0.000	-0.000	-0.000
	(0.000)	(0.001)	(0.000)	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)
Illiterate	-0.001	-0.081***	-0.014***	-0.023***	-0.034***	-0.022**	-0.053***	-0.036***	-0.007
	(0.001)	(0.018)	(0.004)	(0.005)	(0.008)	(0.010)	(0.012)	(0.010)	(0.006)
Constant	0.003	0.092***	0.019**	0.025***	0.021*	0.026	0.061**	0.058**	0.038
	(0.002)	(0.034)	(0.008)	(0.008)	(0.011)	(0.026)	(0.031)	(0.029)	(0.025)
Mean of dependent variable	0.000	0.036	0.004	0.006	0.004	0.016	0.019	0.015	0.005
Observations	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285	1,285
<i>R</i> -squared	0.025	0.304	0.191	0.106	0.276	0.349	0.095	0.232	0.147
Number of districts	388	388	388	388	388	388	388	388	388
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
District fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Cluster SE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Note: **p*<0.1, ***p*<0.05, ****p*<0.01. SE (in parentheses) are clustered at the district level. Sectors have been divided following the ISIC 3.1 classification's division in sections. Shares of male workers in the service sectors are expressed as the number of employed persons in each sector over the total number of employed men in working age (15–64 years) in each district.